\begin{table}%ta1
%%\centering

\par
\caption{\label{tab1app}Kinematic and morphological overview.}
\tiny{\begin{tabular}{llcccll}
\hline\hline\noalign{\smallskip}
\multicolumn{1}{c}{Ident}& \multicolumn{1}{l}{RSA}
& \multicolumn{1}{c}{$\epsilon$}
& \multicolumn{1}{c}{($V/\sigma$)$_{\rm scaled}$}
& \multicolumn{1}{c}{Rot.} & \multicolumn{1}{l}{Gas vs. stars kinematic}
& \multicolumn{1}{l}{Gas vs. stars morphological} \\
\multicolumn{1}{c}{}& \multicolumn{1}{c}{}
& \multicolumn{1}{c}{}& \multicolumn{1}{c}{}
& \multicolumn{1}{c}{class}& \multicolumn{1}{l}{peculiarities}
& \multicolumn{1}{l}{ peculiarities}\\
\hline
  NGC~128  & S02(8) pec      & 0.67 & 0.44  &  F   &  CR g-s (1)  &  g- 
d and g-maj $t \approx 23^\circ$ (5)\\
  NGC~777  & E1             & 0.21 & 0.06  &  S   &               &  \\
  NGC~1052 & E3/S0         & 0.28 & 0.27 &  F   &  CR g-g (1,5)   & \\
  NGC~1209 & E6             & 0.52 & 0.50  &  F   &               & X- 
like struct.; NW linear feature (2) \\
  NGC~1297 & S02/3(0)    & 0.13 &      &     &  &\\
  NGC~1366 & E7/S01(7)    & 0.56 &       &     &  &\\
  NGC~1380 & S03(7)/Sa     & 0.41 & 0.59  & F    &  &\\
  NGC~1389 & S01(5)/SB01  & 0.37 &      &     &  &\\
  NGC~1407 & E0/S01(0)      & 0.07 & 0.11  &  S    & rotat.  min. axis  
(5) &\\
  NGC~1426 & E4             & 0.34 & 0.40  & F    &  &\\
 % & & & &  & &\\
  NGC~1453 & E0             & 0.17 &       &     &  & g-d and g-maj t  
(5) \\
  NGC~1521 & E3             & 0.35 & 0.54   &  F   &  & \\
  NGC~1533 & SB02(2)/SBa   & 0.19 &      &     &  & \\
  NGC~1553 & S01/2(5)pec    & 0.38 & 0.68  & F    &  & shells (3) \\
  NGC~1947 & S03(0) pec     & 0.11 & 0.23  & F     &rotat.  min. axis  
(5)  & dust-lane  min. axis (5)\\
  NGC~2749 & E3             & 0.07 & 0.21 &  F   & rotat.  min. axis  
(5) & \\
  NGC~2911 & S0p or S03(2)  & 0.32 &    &     &  & \\
  NGC~2962 & RSB02/Sa       & 0.37 &  0.51  &  F   &  & \\
  NGC~2974 & E4             & 0.37 & 0.70 & F    & & shells (2); g-d  
and g-maj $t \approx 20^\circ$ (5) \\
  NGC~3136 & E4             & 0.24 &  0.67  &  F   & CR s-s (5) & \\
 % & & & & & &\\
  NGC~3258 & E1             & 0.13 & 0.08 & S    & CR g-s (1)  &\\
  NGC~3268 & E2            & 0.24 & 0.53    & F    &  &\\
  NGC~3489 & S03/Sa     & 0.37 & 0.29 & F    & gas rot. min. axis (5) & 
\\
  NGC~3557 & E3             & 0.21 &  0.30 & F    &  & SW fan; asym.  
outer isophotes (2)\\
  NGC~3607 & S03(3)       & 0.11 & 0.28  &  F   &  &\\
  NGC~3818 & E5             & 0.36 & 0.27 &  F   &  &\\
  NGC~3962 & E1             & 0.22 & 0.23 &  F   &  &gas disk+outer  
arc-like struct. (5)\\
  NGC~4374 & E1              & 0.13 & 0.03 &  S   &  &\\
  NGC~4552 & S01(0)      & 0.06 & 0.05 &  S   &  & shells (3) \\
  NGC~4636 & E0/S01(6)     & 0.24 & 0.07 &  S  & gas irr. motion (5)  & 
\\
 % & & & & &\\
  NGC~4696 &(E3)           & 0.34 &  0.18  & F    & &  Faint outer  
shells (2) \\
  NGC~4697 & E6             & 0.32 &  0.39  &F     &  & Non spherical  
inner isophotes (2) \\
  NGC~5011 & E2             & 0.15 & 0.11  &S     & &\\
  NGC~5044 & E0             & 0.11 & 0.13  &F     & CR s-s; gas  
irr.motion (5)  & gas fil. shape (5)\\
  NGC~5077 & S01/2(4)   & 0.15 & 0.06  &   S  &  CR s-s (5);  g-d and  
g-maj $t \approx 90^\circ$ (5) &\\
  NGC~5090 & E2             & 0.15 & 0.17  & F    & & \\
  NGC~5193 & S01(0)       & 0.07 &    &     &  &\\
  NGC~5266 & S03(5) pec    & 0.31 & 0.12  & S &  stellar rot. min.  
axis (5)   & dust-lane and gas along min. axis (5) \\
  NGC~5328 & E4             & 0.31 &    &     & & \\
  NGC~5363 & [S03(5)]    & 0.34 & 0.39   & F    &  &  dust-lane along  
min. axis (5)\\
 %  & & & & & &\\
  NGC~5638 & E1              & 0.11 &  0.21   &  F   & & \\
  NGC~5812 & E0              & 0.13 & 0.11    & S    & & Tidal tail  
(2) \\
  NGC~5813 & E1              & 0.15 & 0.14    &  S   & CR s-s; gas  
irr. motion (5) & gas fil. shape (5)  \\
  NGC~5831 & E4             & 0.15 & 0.08    &  S   &  CR s-s (6,7) & \\
  NGC~5846 & S01(0)      & 0.07 & 0.03   & S    & gas irr. motion (5)   
& Faint outer shells (2) \\
  NGC~5898 & S02/3(0)   & 0.07 &  0.21 & F    &  CR g-s (1) &  Three  
spiral arm-like tidal tails (2) \\
  NGC~6721 & E1              & 0.15 &  0.26   &  F   & & \\
  NGC~6758 & E2 (merger)    & 0.22 & 0.12    & S    & CR g-s (5) & \\
  NGC~6776 & E1 pec         & 0.17 &  0.24   & F    & &  shell or  
loops NE (3,4); gas fil.  shape (5) \\
  NGC~6868 & E3/S02/3(3)    & 0.19 & 0.14 & F    & CR g-s (5)  & \\
 %  & & & & & &\\
NGC~6875 & S0/a(merger)    & 0.41  &    &     & & \\
NGC~6876 & E3              & 0.13  &    &     &  & \\
NGC~6958 & R?S01(3)   & 0.15  & 0.20   & F    & & Shells (2,3)\\
NGC~7007 & S02/3/a       & 0.42  &    &     & CR g-s (1) & g-d and g- 
maj $t \approx 30^\circ$ (5) \\
NGC~7079  & SBa            & 0.32  & 0.96 &  F& CR g-s + s-s (1) &  \\
NGC~7097 & E4              & 0.29  & 0.10   & S    & CR s-s (1) & \\
NGC~7135 & S01 pec       & 0.31  &0.20    & F    &  & jet-like debris  
and shells (3) \\
NGC~7192 & S02(0)         & 0.15  &  0.04  & S    & CR s-s (5) &   
Shell (2) \\
NGC~7332 & S02/3(8)      & 0.42  & 0.32   & F & CR g-g (1) & \\
NGC~7377 & S02/3/Sa pec   & 0.19  &    &     & & \\
%& & & & & &\\
IC~1459    & E4            & 0.28  &  0.10  &  S   & CR g-g (1) &  
shells (2) \\
IC~2006    & E1            & 0.15  &  0.16  &  F   & CR g-s (1) &  \\
IC~3370    & E2 pec    & 0.21  & 0.28  &  F   &  &  X-like struct.;  
broad N fan (2); polar ring ? (5) \\
IC~4296    & E0           & 0.17  & 0.13   &  F   &  CR s-s (5)  & \\
IC~5063    & S03(3)pec/Sa  & 0.28  & 0.26  & F    &  &\\
%& & & & & &\\
\hline
\end{tabular}}
\tablefoot{Columns.~4 and 5 list respectively (V/$\sigma)_{\rm scaled} =0.57$(V/$\sigma)_{\rm slit}$, and the rotation class (F~=~fast rotator; S~=~slow rotator).  Legenda: {\bf CR g-s}: counter rotation gas vs. stars; {\bf CR s-s}: counter rotation stars vs. stars; {\bf CR g-g}: counter rotation gas vs. gas; {\bf stars rotat.  min.  axis}: stars rotate along the galaxy minor axis; {\bf g-d and g-maj t}: gas disk and galaxy major  
axis are tilted by the reported angle, if provided in the literature.}\vspace{-2mm}
 \tablebib{(1) Corsini \& Bertola \cite{Corsini98}; (2) Tal et~al. \cite{Tal09}; (3) Malin \& Carter \cite{MC83};  (4) Pierfederici \& Rampazzo \cite{Pierfederici04}; (5) the description of the kinematic and morphological peculiarities of the galaxies and full references are reported in the on-line notes of Paper~I and  Paper~II; (6)~Davies et~al. \cite{Davies83}; (7)~Emsellem et~al. \cite{Emsellem04}.}
\end{table}